Author’s response to reviews

Title: A decision tool to help researchers make decisions about including systematic reviews in overviews of reviews of healthcare interventions

Authors:
Michelle Pollock (michelle.pollock@ualberta.ca)
Ricardo Fernandes (rmfernandes@campus.ul.pt)
Amanda Newton (an6@ualberta.ca)
Shannon Scott (ss14@ualberta.ca)
Lisa Hartling (hartling@ualberta.ca)

Version: 1 Date: 25 May 2018

Author’s response to reviews:

21-May-2018

Dear Dr. Estcourt,

Re: Manuscript SYSR-D-18-00038.

As per editorial request we have attached a clean copy of the manuscript; however, we have not included a PRISMA checklist or flow diagram as this reporting guideline is not appropriate for methods studies such as ours.

Sincerely,

Lisa Hartling